Author's response to reviews

Title: Adenovirus type 7 associated with severe and fatal acute lower respiratory infections in argentine children

Authors:

Guiadalupe Carballal (gcarballal@cemic.edu.ar)
Cristina Videla (cvidela@cemic.edu.ar)
Alicia Misirlian (aliciamisir@arnet.com.ar)
Paula Requeijo (prequeijo@hotmail.com)
Maria del Carmen Aguilar (mcaguilar@hotmail.com)

Version: 6 Date: 12 Aug 2002

Changes were made in the title, affiliations and abstract as suggested the Editorial Assistant.